\documentclass[12pt,reqno]{article}

\usepackage{amsmath}
\usepackage{amssymb}
\usepackage{amsthm}
\usepackage{epsf}
\usepackage{float}

\usepackage[usenames]{color}

\usepackage[colorlinks=true,linkcolor=webgreen, filecolor=webbrown,citecolor=webgreen]{hyperref}

\definecolor{webgreen}{rgb}{0,.5,0}
\definecolor{webbrown}{rgb}{.6,0,0}

\newcommand{\seqnum}[1]{\href{http://oeis.org/#1}{\underline{#1}}}


\theoremstyle{plain}
\newtheorem{thm}{Theorem}
\newtheorem{thmm}{Theorem}
\renewcommand{\thethmm}{}
\newtheorem{lem}{Lemma}
\newtheorem{cor}{Corollary}
\newtheorem{claim}{Claim}[section]
\newtheorem{prop}{Proposition}[section]
\theoremstyle{definition}

\newtheorem{defn}{Definition}[section]
\newtheorem{conj}{Conjecture}
\newtheorem{conja}{Conjecture\!}
\renewcommand{\theconja}{}
\newtheorem{exmp}{Example}[section]
\newtheorem{prob}{Problem\!\!}
\renewcommand{\theprob}{}

\theoremstyle{remark}

\newtheorem{rem}{Remark}
\numberwithin{equation}{section}

%%%%%%   General macros  %%%%%%%%
%
% aeq   approximately equal
        \newcommand{\aeq}[3]{$\,{#1}\,
                          {\overset{#2}{=}}\,{#3}\,$}
% aeqc  approximately equal centreed
        \newcommand{\aeqc}[3]{\begin{equation}{#1}\,
                      {\overset{#2}{=}}\,{#3}\end{equation}}
% aeqn  approximately equal numbered
        \newcommand{\aeqn}[3]{\begin{equation}{#1}\,
                   {\overset{#2}{=}}\,{#3} \end{equation}}
% aeqw   approximately equal
        \newcommand{\aeqw}[3]{{#1}\,
                       {\overset{#2}{=}}\,{#3}}
% ahu  approximately upper homotopic
        \newcommand{\ahu}[3]{${#1}\,
                              \overset{#2}{\simeq}\,{#3}\,$}
% ahw  approximately upper homotopic
        \newcommand{\ahw}[3]{\,{#1}\,
                                 \overset{#2}{\simeq}\,{#3}\,}
% ahuc  approximately upper homotopic centreed
        \newcommand{\ahuc}[3]{\begin{equation}{#1}\,
                   \overset{#2}{\simeq}\,{#3}\end{equation}}
% ahun  approximately equal numbered
        \newcommand{\ahun}[3]{\begin{equation}{#1}\,
                 \overset{#2}{\simeq}\,{#3}\end{equation}}
% ang
    \newcommand{\ang}[3]{\langle {#1},\,{#2},\,{#3}\rangle}

% aph  approximately proper homotopic
        \newcommand{\aph}[3]{${#1}\,
                           \overset{#2}{{\simeq}_p}\,{#3}\,$}
% aphw  approximately proper homotopic
        \newcommand{\aphw}[3]{\,{#1}\,
                             \overset{#2}{{\simeq}_p}\,{#3}\,}
% aphc  approximately upper homotopic centreed
        \newcommand{\aphc}[3]{\begin{equation}{#1}\,
              \overset{#2}{{\simeq}_p}\,{#3}\end{equation}}
% aphn  approximately proper {homotopic} numbered
        \newcommand{\aphn}[3]{\begin{equation}{#1}\,
               \overset{#2}{{\simeq}_p}\,{#3}\end{equation}}
% bo    bold
        \newcommand{\bo}[1]{\bold {#1}}
% bz    bold
        \newcommand{\bz}[1]{\z {\bold {#1}}}
% cd    composition of two
        \newcommand{\cd}[2]{{#1}\circ{#2}}
% cf    composition of four
        \newcommand{\cf}[4]{{#1}\circ{#2}\circ{#3}\circ{#4}}
% chu   check on uppercase
        \newcommand{\chu}[1]{\check {\up {#1}}}
% ct    composition of three
        \newcommand{\ct}[3]{{#1}\circ{#2}\circ{#3}}
% e      element of
        \newcommand{\e}[2]{{#1}\in {#2}}
% ed    element of "od dva"
        \newcommand{\ed}[4]{{#1}\in {\odd {#2}{#3}{#4}}}
% edz    element of "od dva"
        \newcommand{\edz}[4]{${#1}\in {\odd {#2}{#3}{#4}}$}
% ef    element of class up f of functions from up x to up y
        \newcommand{\ef}[4]{$#1\in {\cal {\up
                                  {#2}}}_{\up {#3}}^{\up {#4}}$}
% eh    element of h
        \newcommand{\eh}[2]{{#1}\in {\Hat {#2}}}
% ehz   element of h
        \newcommand{\ehz}[2]{${#1}\in {Cov({#2})}$}
% ehuz  element of hu
        \newcommand{\ehuz}[2]{\ez {#1}{\hu{#2}}}
% ek    element of k
%        \newcommand{\ek}[2]{{#1}\in{\k {#2}}}
% ekz   element of k
        \newcommand{\ekz}[2]{${#1}\in{\cal {\up {#2}}}$}
% el    exists
        \newcommand{\el}{\exists}
% eq    equal
        \newcommand{\eq}[2]{{#1} = {#2}}
% eqc   equal centreed
        \newcommand{\eqc}[2]{\begin{equation}{#1} = {#2}\end{equation}}
% eqz   equal
        \newcommand{\eqz}[2]{${#1} = {#2}$}
% essz  element of something with double star
        \newcommand{\essz}[2]{${#1}\in{#2}^{\star\star}$}
% esz   element of something with star
        \newcommand{\esz}[2]{${#1}\in {#2}^{\star}$}
% et    element of t (tilde)
        \newcommand{\et}[2]{{#1}\in {Inc({#2})}}
% etz   element of t
        \newcommand{\etz}[2]{${#1}\in {Inc({#2})}$}
% euz    element of up
        \newcommand{\euz}[2]{${#1}\in {\up {#2}}$}
% ez    element of
        \newcommand{\ez}[2]{${#1}\in {#2}$}
% f    function
        \newcommand{\f}[3]{{#1}:{#2}\rightarrow{#3}}
% fl    forall
        \newcommand{\fl}{\forall}
% fz    function
        \newcommand{\fz}[3]{${#1}:{#2}\rightarrow{#3}$}
% fw    function
        \newcommand{\fw}[3]{{#1}:{#2}\rightarrow{#3}}

% fuz    function of up
        \newcommand{\fuz}[3]{${#1}:{\up {#2}}\rightarrow
                                                       {\up {#3}}$}
% gee   greater or equal
        \newcommand{\gee}[2]{$#1>#2$}
% got    greek-overline-text
       \newcommand{\got}[2]{\z {\overline {#1}}-{#2}}
% gsm  greek-small 1-matrix
        \newcommand{\gsm}[3]{\z {#1}--\sm {#2}{#3}}
% gt      greek-text
        \newcommand{\gt}[2]{${#1}$-{#2}}
% h     hat on
        \newcommand{\h}[1]{\Hat {#1}}
% hl    half line skip
        \newcommand{\hl}{\vskip 0.5\baselineskip\par}
% ho    homotopic
        \newcommand{\ho}[2]{{#1}\simeq{#2}}
% hoc   homotopic centreed
        \newcommand{\hoc}[2]{\begin{equation}{#1}\,
                                      \simeq\, {#2}\end{equation}}
% hon   homotopic centreed and numbered
        \newcommand{\hon}[2]{\begin{equation}{#1}\,
                  \simeq\, {#2}\end{equation}}
% how   homotopic
        \newcommand{\how}[2]{{#1}\simeq {#2}}
% hoz   homotopic
        \newcommand{\hoz}[2]{${#1}\simeq {#2}$}
% hu    hat on uppercase
        \newcommand{\hu}[1]{\Hat {\up {#1}}}
% huz   hat on uppercase
        \newcommand{\huz}[1]{$\Hat {\up {#1}}$}
% hz    hat on
        \newcommand{\hz}[1]{$\Hat {#1}$}
% hw    hat on
        \newcommand{\hw}[1]{\Hat {#1}}
% iw    inclusion
        \newcommand{\iw}[2]{i_{{\up {#1}}, {\up {#2}}}}
% iz    inclusion
        \newcommand{\iz}[2]{$\, i_{{\up {#1}}, {\up {#2}}}\,$}
% k     caligraphic
%        \newcommand{\k}[1]{\cal {\up {#1}}}
% kz    caligraphic
        \newcommand{\kz}[1]{$\cal {\up {#1}}$}
% lee   less than or equal
        \newcommand{\lee}[2]{$\,{#1}\,\le\,{#2}\,$}
% nst   n-th star (of a cover)
        \newcommand{\nstw}[2]{st^{#1}({#2})}
% nstz  n-th star of (a cover)
        \newcommand{\nst}[2]{$st^{#1}({#2})$}
% od    "od"
        \newcommand{\od}[2]{{#1}({#2})}
% odd   " od dva"
        \newcommand{\odd}[3]{{#1}({#2},\, {#3})}
% oddz   " od dva"
        \newcommand{\oddz}[3]{${#1}({#2},\, {#3})$}
% odz   "od"
        \newcommand{\odz}[2]{${#1}({#2})$}
% oz   overline
       \newcommand{\oz}[1]{$\overline {#1}$}
% ow   overline
       \newcommand{\ow}[1]{\overline {#1}}
% p     prime
        \newcommand{\p}[1]{{#1}^{\prime}}
% pa     perspective
        \newcommand{\pa}{{\,{\langle\rangle}\,}}
% pe     perspective from a point
        \newcommand{\pe}[1]{{\,{\langle{#1}\rangle\,}}}
% prz    pair (x,  y)}
        \newcommand{\prz}[2]{$({#1},\,{#2})$}
% pru    pair (up x, up y)}
        \newcommand{\pru}[2]{$(\up {#1},\,\up {#2})$}
% prv    pair (up x, up y)} without
        \newcommand{\prv}[2]{(\up {#1},\,\up {#2})}
% prw    pair (x,  y)} without
        \newcommand{\prw}[2]{({#1},\,{#2})}
% pz    prime
        \newcommand{\pz}[1]{\z {\p {#1}}}
% qr  quadruple with ; at the middle
        \newcommand{\qr}[4]{$({#1},\,{#2};\,{#3},\,{#4})$}
% qrw  quadruple with ; at the middle
        \newcommand{\qrw}[4]{({#1},\,{#2};\,{#3},\,{#4})}
% qz    p with  index xy on -1(z)}
        \newcommand{\qz}[4]{${#1}_{#2#3}^{-1}(#4)$}
% re     reference
        \newcommand{\re}[1]{\cite{#1}}
% rev    restriction to uppercase
        \newcommand{\rev}[2]{{#1\vert}_{\up {#2}}}
% rez   restriction to uppercase
        \newcommand{\rez}[2]{$\,{#1}{\vert}_{\up{#2}}\,$}
% rk    reference with comment
        \newcommand{\rk}[2]{\cite[#2]{#1}}
% s     set
        \newcommand{\s}[3]{\{ {#1}\, :\>{#2}\;\in\,{#3}\}}
% sh    S-homotopic
        \newcommand{\sh}[2]{{#1}\,{\simeq}_S\,{#2}}
% shc   S-homotopic centreed
        \newcommand{\shc}[2]{\begin{equation}{#1}\,
                                      {\simeq}_S\, {#2}\end{equation}}
% shn   S-homotopic centreed and numbered
        \newcommand{\shn}[2]{\begin{equation}{#1}\,
                  {\simeq}_S\, {#2}\end{equation}}
% shw   S-homotopic
        \newcommand{\shw}[2]{{#1}\,{\simeq}_S\,{#2}}
% shz   S-homotopic
        \newcommand{\shz}[2]{${#1}\,{\simeq}_S\,{#2}$}
% st    star
        \newcommand{\st}[2]{st({#1}, {#2})}
% stc   star of a cover
        \newcommand{\stc}[1]{{st({#1})}}
% stcz  star of a cover
        \newcommand{\stcz}[1]{${st({#1})}$}
% stz   star
        \newcommand{\stz}[2]{$st({#1},\, {#2})$}
% suz    subset
       \newcommand{\suz}[2]{${#1}\subset {#2}$}
% suw    subset
       \newcommand{\suw}[2]{{#1}\subset {#2}}
% sz    set
        \newcommand{\sz}[3]{$\{{#1}:\medspace{#2}\in
                                                {#3}\}$}
% tr    triple
        \newcommand{\tr}[3]{$({#1},\,{#2},\,{#3})$}
% trw    triple
      \newcommand{\trw}[3]{({#1},\,{#2},\,{#3})}
% tw    tilda
        \newcommand{\tw}[1]{{\widetilde {#1}}}
% tz   tilda
        \newcommand{\tz}[1]{${\widetilde {#1}}$}
% up    uppercase letters
        \newcommand{\up}{\uppercase}
% uz    underlined
        \newcommand{\uz}[1]{$\underline {#1}$}
% uw    underlined
        \newcommand{\uw}[1]{\underline {#1}}
% w     indexed up
        \newcommand{\w}[2]{{#1}^{#2}}
% wz    indexed up
        \newcommand{\wz}[2]{${#1}^{#2}$}
% x     indexed down
        \newcommand{\x}[2]{{#1}_{#2}}
% xz    indexed down
        \newcommand{\xz}[2]{${#1}_{#2}$}
% xx    indexed down by two indices xy
        \newcommand{\xx}[3]{{#1}_{{#2}{#3}}}
% xxz   indexed by two indices xy
        \newcommand{\xxz}[3]{$\,{#1}_{{#2}{#3}}\,$}
% y     indexed up and down
        \newcommand{\y}[3]{{#1}^{#2}_{#3}}
% yz    indexed up and down
        \newcommand{\yz}[3]{${#1}^{#2}_{#3}$}
% z     math mode
        \newcommand{\z}[1]{${#1}$}
% zj     math mode uppercase
        \newcommand{\zj}[1]{$\up {#1}$}
% q    display mode
        \newcommand{\q}[1]{\[{#1}\]}
% qq    display mode
        \newcommand{\qq}[1]{\begin{equation}{#1}
        \end{equation}}
% zv   star
       \newcommand{\zv}[1]{{#1}^{*}}
%
%%%%    MACROS FOR GREEK LETTERS
%
% alpha                                   -> A
         \newcommand{\za}{\alpha}
% beta                                    -> B
         \newcommand{\zb}{\beta}
% psi                                     -> C
          \newcommand{\zc}{\psi}
          \newcommand{\zC}{\Psi}
          \newcommand{\vC}{\varPsi}
% delta                                   -> D
          \newcommand{\zd}{\delta}
          \newcommand{\zD}{\Delta}
          \newcommand{\vD}{\varDelta}
% epsilon                                 -> E
          \newcommand{\ze}{\epsilon}
          \newcommand{\ve}{\varepsilon}
% phi                                     -> F
         \newcommand{\zf}{\phi}
          \newcommand{\zF}{\Phi}
          \newcommand{\vf}{\varphi}
          \newcommand{\vF}{\varPhi}
% gamma                                  -> G
          \newcommand{\zg}{\gamma}
          \newcommand{\zG}{\Gamma}
          \newcommand{\vG}{\varGamma}
% eta                                    -> H
          \newcommand{\zh}{\eta}
% iota                                   -> I
          \newcommand{\zi}{\iota}
%                                        -> J  (unused)
% kappa                                  -> K
          \newcommand{\zk}{\kappa}
% lambda                                 -> L
          \newcommand{\zl}{\lambda}
          \newcommand{\zL}{\Lambda}
          \newcommand{\vL}{\varLambda}
% mu                                    -> M
          \newcommand{\zm}{\mu}
% nu                                    -> N
          \newcommand{\zn}{\nu}
%                                       -> O   (to omicron)
% pi                                    -> P
          \newcommand{\zp}{\pi}
          \newcommand{\zP}{\Pi}
          \newcommand{\vp}{\varpi}
          \newcommand{\vP}{\varPi}
% chi                                  -> Q
          \newcommand{\zq}{\chi}
% rho                                  -> R
          \newcommand{\zr}{\rho}
          \newcommand{\vr}{\varrho}
% sigma                               -> S
          \newcommand{\zs}{\sigma}
          \newcommand{\zS}{\Sigma}
          \newcommand{\vs}{\varsigma}
          \newcommand{\vS}{\varSigma}
% tau                                 -> T
          \newcommand{\zt}{\tau}
% upsilon                             -> U
          \newcommand{\zu}{\upsilon}
          \newcommand{\zU}{\Upsilon}
%                                     -> V  (unused)
% omega                               -> W
          \newcommand{\zw}{\omega}
          \newcommand{\zW}{\Omega}
          \newcommand{\vW}{\varOmega}
% xi                                  -> X
          \newcommand{\zx}{\xi}
          \newcommand{\zX}{\Xi}
          \newcommand{\vX}{\varXi}
% theta                               -> Y
          \newcommand{\zy}{\theta}
          \newcommand{\zY}{\Theta}
          \newcommand{\vy}{\vartheta}
          \newcommand{\vY}{\varTheta}
% zeta                                -> Z
          \newcommand{\zz}{\zeta}
%%%%%%     END OF MACROS     %%%%%
\hyphenation{app-ro-xi-ma-te to-po-lo-gy
pro-per-ly pro-per-ties cor-res-pond
a-na-lo-go-us equi-va-len-ce equi-va-len-ces
ho-mo-lo-go-us tri-li-ne-ar e-qui-la-te-ral
tri-ang-les Theo-rem ge-ne-ra-li-ty cong-ruent
Ma-te-ma-ti-co} \hfuzz=5pt

\setlength{\textwidth}{6.5in}
\setlength{\oddsidemargin}{.1in}
\setlength{\evensidemargin}{.1in}
\setlength{\topmargin}{-.5in}
\setlength{\textheight}{8.9in}

\begin{document}
\begin{thebibliography}{99}
%
\bibitem{c1-D}
Z. \v{C}erin, Properties of odd and even
terms of the Fibonacci sequence, {\it Demonstratio
Math.}, {\bf 39} (2006), 55--60.
%
%
\bibitem{c2-B}
Z. \v{C}erin, On sums of squares of odd
and even terms of the Lucas sequence,
{\it Proccedings of the 11th Fibonacci Conference},
to appear.
%
%
\bibitem{c3-P}
Z. \v{C}erin, Some alternating sums of
Lucas numbers, {\it Central European J. 
Math.} {\bf 3} (2005), 1--13.
%
%
\bibitem{c4-M}
Z. \v{C}erin, Alternating sums of
Fibonacci products, {\it Atti del Seminario
Matematico e Fisico dell'Universit\`{a} di
Modena e Reggio Emilia}, (to appear).
%
\vspace{5pt}
%
\bibitem{c5-PL}
Z. \v{C}erin and G. M. Gianella, On sums
of squares of Pell-Lucas numbers, {\it INTEGERS:
Electronic Journal of Combinatorial Number
Theory}, {\bf 6} (2006), \# A15.  Available at
\href{http://www.integers-ejcnt.org/vol6.html}{\tt http://www.integers-ejcnt.org/vol6.html}.
%
%
\bibitem{c6-P}
Z. \v{C}erin and G. M. Gianella, Formulas
for sums of squares and products of Pell
numbers, {\it Acc. Sc. Torino - Atti Sci. Fis.}
{\bf 140} (2006), to appear.
%
%
%\bibitem{hog}
%V. E. Hoggatt, Jr., {\em Fibonacci and Lucas
%numbers}, The Fibonacci Association, Santa
%Clara, 1969.
%
%
\bibitem{hor}
A. F. Horadam,  Jacobsthal representation
numbers, {\it Fib. Quart.} {\bf 34} (1996), 40--54.
%
%\bibitem{kn}
%R. Knott, {\em Fibonacci numbers and the
%Golden Section},\\
%\verb"http://www.mcs.surrey.ac.uk/Personal/R.Knott/Fibonacci/fib.html".
%
%
\bibitem{r&l}
V. Rajesh and G. Leversha, Some
properties of odd terms of the Fibonacci
sequence, {\it Math. Gazette}, {\bf 88}
(2004), 85--86.
%
\bibitem{sloane}
N.~J.~A. Sloane,
\htmladdnormallink{\textit{The On-Line
Encyclopedia of Integer Sequences}}
{http://www.research.att.com/$\sim$njas/sequences/},
2006.

%
%\bibitem{sloane}
%N. J. A. Sloane, {\em On-Line Encyclopedia of
%Integer Sequences},\\
%\verb"http://www.research.att.com/~njas/sequences/".
%
%
%\bibitem{vajda}
%S. Vajda, {\em Fibonacci and Lucas numbers,
%and the Golden Section: Theory and
%Applications}, Halsted Press, Chichester
%(1989).
%
\end{thebibliography}

\begin{center}
\epsfxsize=4in \leavevmode\epsffile{logo129.eps}
\end{center}

\begin{center}
\vskip 1cm{\LARGE\bf Sums of Squares
and Products \\
\vskip .1in of Jacobsthal Numbers} \vskip 1cm
\large
Zvonko \v Cerin \\
Department of Mathematics \\
University of Zagreb\\
Bijeni\v{c}ka 30 \\
Zagreb 10 000 \\
Croatia \\
\href{mailto:cerin@math.hr}{\tt cerin@math.hr} \\
\end{center}


\vskip .2in

\begin{abstract}
We consider sums of squares of odd and even
terms of the Jacobsthal sequence and sums of
their products. We also study the analogous alternating sums.
These sums are related to 
products of appropriate Jacobsthal numbers and
several integer sequences. The formulas that
we discover show that a certain translation
property for these sums holds, so that in
practice, only sums of initial values 
and the information where the
summation begins are necessary.
\end{abstract}


\vskip .2in


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%               %%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%   Section 1   %%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%               %%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%\keywords {Jacobsthal
%numbers, Jacobsthal-Lucas numbers, integer
%sequences, sum of squares, combinatorial
%identity, Maple V} \subjclass {Primary 11B39,
%11Y55, 05A19}

\section{Introduction}
The Jacobsthal and Jacobsthal-Lucas sequences
\z {J_n} and \z {j_n} are defined by the
recurrence relations \q {J_0=0,\qquad
J_1=1,\qquad J_n=J_{n-1}+2\,J_{n-2}\,\,\text{
for \z {n\geqslant 2},}} and \q {j_0=2,\qquad
j_1=1,\qquad j_n=j_{n-1}+2\,j_{n-2}\,\,\text{
for \z {n\geqslant 2}.}}

In Sections 2--4 we consider sums of
squares of odd and even terms of the
Jacobsthal sequence and sums of their
products. These sums have nice representations
as products of appropriate Jacobsthal and
Jacobsthal-Lucas numbers.


The numbers \z {J_k} appear as the integer
sequence \z {A001045} from \re {sloane} while
the numbers \z {j_k} is \z {A014551}. The
properties of these numbers are summarized in
\re {hor}. For the convenience of the reader
we shall now explicitly define these
sequences.

The first eleven terms of the sequence \z
{J_k} are \z {0}, \z {1}, \z {1}, \z {3}, \z
{5}, \z {11}, \z {21}, \z {43}, \z {85}, \z
{171} and \z {341}. It is given by the formula
\z {J_n={\frac{2^n-(-1)^n}{3}}}.

The first eleven terms of the sequence \z
{j_k} are \z {2}, \z {1}, \z {5}, \z {7}, \z
{17}, \z {31}, \z {65}, \z {127}, \z {257}, \z
{511} and \z {1025}. It is given by the
formula \z {j_n=2^n+(-1)^n}.

In the last three sections we look into the
alternating sums of squares of odd and even
terms of the Jacobsthal sequence and the
alternating sums of products of two
consecutive Jacobsthal numbers. These sums
also have nice representations as products of
appropriate Jacobsthal and Jacobsthal-Lucas
numbers.

These formulas for ordinary sums and for
alternating sums have been discovered with the
help of a PC computer and all algebraic
identities needed for the verification of our
theorems can be easily checked in either
Derive, Mathematica or Maple V. Running times
of all these calculations are in the range of
a few seconds.

Similar results for Fibonacci, Lucas, Pell,
and Pell-Lucas numbers have recently been
discovered by G. M. Gianella and the author in
papers \cite{c1-D,c2-B,c3-P,c4-M,c5-PL,c6-P}. They
improved some results in \cite{r&l}.

\section{Jacobsthal even squares}

The following lemma is needed to accomplish
the inductive step in the proof of the first
part of our first theorem.

For any \z {n=0,\,1,\,2,\dots} let \z
{{\beta}_n={\frac{2^{4n+2}-1}{3}}=J_{4n+2}},
\z
{{\gamma}_n={\frac{2^{4n+2}+1}{5}}={\frac{j_{4n+2}}{5}}}
and \z
{{\tau}_n={\frac{2^{4n}-1}{15}}={\frac{J_{4n}}{5}}}.


\begin{lem}\label{l1}
For every \z {m\geqslant 0} and \z {k\geqslant
0} the following equality holds
\begin{multline}
 \hspace{2cm}   ({\beta}_{n+1}\,{\gamma}_{n+1}-
{\beta}_{n}\,{\gamma}_{n})\,J_{2k}^2+
8\,({\beta}_{n+1}\,{\tau}_{n+1}-
{\beta}_{n}\,{\tau}_{n})\,J_{2k}\\
    =J_{2k+4n+4}^2+J_{2k+4n+2}^2-
J_{4n+4}^2-J_{4n+2}^2.\hspace{2cm}
%\quad\text{\rm (2.1)}
\end{multline}
\end{lem}
\begin{proof} ({\textbf{P1}}).
Let \z {A=2^{2k}}, \z {B=2^{8n}} and \z
{C=2^{4n}}. The difference of the left hand
side and the right hand side of the relation
(2.1) is equal to
\begin{multline*}
{\frac {544}{9}}\,\left(  \left( -1 \right)
^{1+2\,k}+1 \right) AB+ {\frac {40}{9}}\,
\left( \left( -1 \right) ^{4\,n+2\,k}+  \left(
-1 \right) ^{1+4\,n} \right) (A+1)C\\+ { \frac
{272}{9}}\,\left( \left( -1 \right)
^{4\,k}+2\, \left( -1 \right)
^{1+2\,k}+1\right) B+{\frac{2}{9}}\,\left(
 \left( -1 \right) ^{8\,n}+
 \left( -1 \right) ^{1+8\,n+4\,k}\right).
\end{multline*}
It is obvious that all coefficients in the
above expression are zero and the proof is
complete.
\end{proof}

There are essentially three types of proofs in
this paper that we indicate as {\textbf{P1}},
{\textbf{P2}} and {\textbf{P3}}.

In {\textbf{P1}} and {\textbf{P2}} we use the
substitutions \z {A=2^{2k}}, \z {B=2^{8n}}, \z
{C=2^{4n}} and \z {M=2^k}, \z {P=(-1)^k},
respectively. We prove that the difference of
the left hand side and the right hand side of
the relation from the statement is equal to
zero with algebraic simplification to an
expression that is obviously vanishing. This
part is easily done with the help of a
computer (for example in Maple V) so that we
shall only indicate the final expression.

In {\textbf{P3}} we argue by induction on \z
{n} taking care first of the initial value \z
{n=0} and then showing that it holds for \z
{n=r+1} under the assumption that it is true
for \z {n=r}.

The following lemma is needed to accomplish
the inductive step in the proof of the second
part of our first theorem.

For any \z {n=0,\,1,\,2,\dots} let \z
{{\eta}_n=2^{4n+4}+1=j_{4n+4}}.

\begin{lem}\label{l2}
For every \z {m\geqslant 0} and \z {k\geqslant
0} the following equality holds
\begin{multline}
\hspace{2cm}({\tau}_{n+2}\,{\eta}_{n+1}-
{\tau}_{n+1}\,{\eta}_{n})\,J_{2k}^2+
8\,({\beta}_{n+1}\,{\tau}_{n+2}-
{\beta}_{n}\,{\tau}_{n+1})\,J_{2k}\\
    =J_{2k+4n+6}^2+J_{2k+4n+4}^2-
J_{4n+6}^2-J_{4n+4}^2.\hspace{2cm}
\end{multline}
\end{lem}
\begin{proof} ({\textbf{P1}}).
\begin{multline*}
{\frac {8704}{9}}\,\left(  \left( -1 \right)
^{1+2\,k}+1 \right) AB+ {\frac {160}{9}}\,
\left( \left( -1 \right) ^{4\,n+2\,k}+  \left(
-1 \right) ^{1+4\,n} \right) (A+1)C\\+ { \frac
{4352}{9}}\,\left( \left( -1 \right)
^{4\,k}+2\, \left( -1 \right)
^{1+2\,k}+1\right) B+{\frac{2}{9}}\,\left(
 \left( -1 \right) ^{8\,n}+
 \left( -1 \right) ^{1+8\,n+4\,k}\right).
\end{multline*}
\end{proof}

\begin{thm}\label{t1}
For every \z {m\geqslant 0} and \z {k\geqslant
0} the following equalities hold \qq
{\sum_{i=0}^{m}\,J_{2k+2i}^2=
\sum_{i=0}^{m}\,J_{2i}^2+{\beta}_n\,J_{2\,k}
\left[{\gamma}_n\,J_{2k}+ 8\,{\tau}_n\right],}
if \z {m=2n} and \z {n=0,\,1,\,2,\dots}  and
\qq { \sum_{i=0}^{m}\,J_{2k+2i}^2=
\sum_{i=0}^{m}\,J_{2i}^2+{\tau}_{n+1}\,J_{2\,k}
\left[{\eta}_n\,J_{2k}+ 8\,{\beta}_n\right],}
if \z {m=2n+1} and \z {n=0,\,1,\,2,\dots}.
\end{thm}
\begin{proof}[Proof of (2.3)] ({\textbf{P3}}).
The proof is by induction on \z {n}. When \z
{n=0} we obtain \q {
J_{2k}^2=J_{0}^2+{\beta}_0\,J_{2k}\,
[{\gamma}_0\,J_{2k}+8\,{\tau}_0]=J_{2k}\,J_{2k}=
J_{2k}^2,} because \z {J_0=0}, \z
{{\beta}_0=1}, \z {{\gamma}_0=1} and \z
{{\tau}_0=0}.

Assume that the relation (2.3) is true for \z
{n=r}. Then
\begin{multline*}
    \sum_{i=0}^{ 2(r+1)}\,J_{2k+2i}^2=
J_{2k+4r+2}^2+J_{2k+4r+4}^2+
\sum_{i=0}^{2r}\,J_{2k+2i}^2=
J_{2k+4r+2}^2+J_{2k+4r+4}^2+\\
\sum_{i=0}^{2r}\,J_{2i}^2+{\beta}_r\,J_{2\,k}
\left[{\gamma}_r\,J_{2k}+ 8\,{\tau}_r\right]
=\sum_{i=0}^{2(r+1)}\,J_{2i}^2+{\beta}_{r+1}\,J_{2\,k}
\left[{\gamma}_{r+1}\,J_{2k}+
8\,{\tau}_{r+1}\right],
\end{multline*}
where the last step uses Lemma 1 for \z
{n=r+1}. Hence, (2.3) is true for \z {n=r+1}
and the proof is completed.
\end{proof}
\begin{proof}[Proof of (2.4)] ({\textbf{P3}}).
When \z {n=0} we obtain \q {
J_{2k}^2+J_{2k+2}^2=J_{0}^2+J_{2}^2+{\tau}_1\,J_{2k}\,
[{\eta}_0\,J_{2k}+8\,{\beta}_0]=
17\,J_{2k}^2+8\,J_{2k}+1,} since \z {J_0=0},
\z {J_1=1}, \z {{\beta}_0=1}, \z {{\eta}_0=17}
and \z {{\tau}_1=1}. The above equality is
equivalent to \z {J_{2k+2}^2=16\,J_{2k}^2+8\,
J_{2k}+1=(4\,J_{2k}+1)^2} which is true
because it follows from the relation \z
{J_{n+2}=4\,J_n+1}.

Assume that the relation (2.4) is true for \z
{n=r}. Then
\begin{multline*}
    \sum_{i=0}^{ 2(r+1)+1}\,J_{2k+2i}^2=
J_{2k+4r+4}^2+J_{2k+4r+6}^2+
\sum_{i=0}^{2r+1}\,J_{2k+2i}^2=
J_{2k+4r+4}^2+J_{2k+4r+6}^2+\\
\sum_{i=0}^{2r+1}\,J_{2i}^2+{\tau}_{r+1}\,J_{2\,k}
\left[{\eta}_r\,J_{2k}+ 8\,{\beta}_r\right]
=\sum_{i=0}^{2(r+1)+1}\,J_{2i}^2+{\tau}_{r+2}\,J_{2\,k}
\left[{\eta}_{r+1}\,J_{2k}+
8\,{\beta}_{r+1}\right],
\end{multline*}
where the last step uses Lemma 2 for \z
{n=r+1}.
\end{proof}

\section{Jacobsthal odd squares}
The initial step in an inductive proof of the
first part of our second theorem uses the
following lemma.
\begin{lem}\label{l3}
    For every \z {k\geqslant 0} the following
    identity holds
    \q {J_{2k+1}^2=
    16\,J_{2k}\,J_{2k-2}+8\,J_{2k}+1.}
\end{lem}
\begin{proof} ({\textbf{P2}}).
Let \z {M=2^k} and \z {P=(-1)^k}. The
difference of the left and the right hand side
is equal to \z {{\frac{\left( P-1 \right)
\left( P+1
\right)}{3}}\left(8\,M^2-5\,{P}^{2}+3
\right)=0.}
\end{proof}
The initial step in an inductive proof of the
second part of our second theorem uses the
following lemma.
\begin{lem}\label{l4}
    For every \z {k\geqslant 0} the following
    identity holds
    \q {J_{2k+1}^2+J_{2k+3}^2=10+
    8\,J_{2k}\,(34\,J_{2k-2}+15).}
\end{lem}
\begin{proof} ({\textbf{P2}}).
\z {10\left( P-1 \right) \left( P+1 \right)
(4\,{M}^{2}- 3\,{P}^{2}+1).}
\end{proof}

The following lemma is needed to accomplish
the inductive step in the proof of the first
part of our second theorem.

For any \z {n=0,\,1,\,2,\dots} let \z
{{\pi}_n={\frac{2^{4n+1}-1}{3}}=J_{4n+1}}.

\begin{lem}\label{l5}
For every \z {m\geqslant 0} and \z {k\geqslant
0} the following equality holds
\begin{multline}
 \hspace{2cm}  8\,J_{2k}\,\left[2\,({\beta}_{n+1}\,
   {\gamma}_{n+1}-{\beta}_{n}\,{\gamma}_{n})\,
   J_{2k-2}+
{\beta}_{n+1}\,{\pi}_{n+1}-
{\beta}_{n}\,{\pi}_{n}\right]\\
    =J_{2k+4n+5}^2+J_{2k+4n+3}^2-
J_{4n+5}^2-J_{4n+3}^2.\hspace{2cm}
\end{multline}
\end{lem}
\begin{proof} ({\textbf{P1}}).
\begin{multline*}
{\frac {5440}{9}}\,\left(  \left( -1 \right)
^{1+2\,k}+1 \right) AB+ {\frac {80}{9}}\,
\left( \left( -1 \right) ^{4\,n+2\,k+1}+
\left( -1 \right) ^{4\,n} \right) (A+1)C\\+ {
\frac {1088}{9}}\,\left( 4\,\left( -1 \right)
^{4\,k}+5\, \left( -1 \right)
^{1+2\,k}+1\right) B+{\frac{2}{9}}\,\left(
 \left( -1 \right) ^{8\,n}+
 \left( -1 \right) ^{8\,n+4\,k+1}\right).
\end{multline*}
\end{proof}
The following lemma is needed to accomplish
the inductive step in the proof of the second
part of our second theorem.

For any \z {n=0,\,1,\,2,\dots} let \z
{{\sigma}_n=5\,J_{4n+3}}.

\begin{lem}\label{l6}
For every \z {m\geqslant 0} and \z {k\geqslant
0} the following equality holds
\begin{multline}
 \hspace{2cm}  8\,J_{2k}\,\left[2\,({\tau}_{n+2}\,
   {\eta}_{n+1}-{\tau}_{n}\,{\eta}_{n})\,
   J_{2k-2}+
{\tau}_{n+2}\,{\sigma}_{n+1}-
{\tau}_{n+1}\,{\sigma}_{n}\right]\\
    =J_{2k+4n+7}^2+J_{2k+4n+5}^2-
J_{4n+7}^2-J_{4n+5}^2.\hspace{2cm}
\end{multline}
\end{lem}
\begin{proof} ({\textbf{P1}}).
\begin{multline*}
{\frac {87040}{9}}\,\left(  \left( -1 \right)
^{1+2\,k}+1 \right) AB+ {\frac {320}{9}}\,
\left( \left( -1 \right) ^{4\,n+2\,k+1}+
\left( -1 \right) ^{4\,n} \right) (A+1)C\\+ {
\frac {17408}{9}}\,\left( 4\,\left( -1 \right)
^{4\,k}+5\, \left( -1 \right)
^{1+2\,k}+1\right) B+{\frac{2}{9}}\,\left(
 \left( -1 \right) ^{8\,n}+
 \left( -1 \right) ^{8\,n+4\,k+1}\right).
\end{multline*}
\end{proof}

\begin{thm}\label{t2}
For every \z {m\geqslant 0} and \z {k\geqslant
0} the following equalities hold \qq
{\sum_{i=0}^{m}\,J_{2k+2i+1}^2=
\sum_{i=0}^{m}\,J_{2i+1}^2+8\,{\beta}_n\,J_{2\,k}
\left[2\,{\gamma}_n\,J_{2k-2}+
{\pi}_n\right],} if \z {m=2n} and \z
{n=0,\,1,\,2,\dots}  and \qq {
\sum_{i=0}^{m}\,J_{2k+2i+1}^2=
\sum_{i=0}^{m}\,J_{2i+1}^2+8\,{\tau}_{n+1}\,J_{2\,k}
\left[2\,{\eta}_n\,J_{2k-2}+{\sigma}_n\right],}
if \z {m=2n+1} and \z {n=0,\,1,\,2,\dots}.
\end{thm}
\begin{proof}[Proof of (3.3)] ({\textbf{P3}}).
When \z {n=0} we obtain \q {
J_{2k+1}^2=J_{1}^2+8\,{\beta}_0\,J_{2k}\,
[2\,{\gamma}_0\,J_{2k-2}+{\pi}_0]=1+8\,J_{2k}\,
[2\,J_{2k-2}+1],} because \z {J_1=1}, \z
{{\beta}_0=1}, \z {{\gamma}_0=1} and \z
{{\pi}_0=1}. But, this equality is true by
Lemma 3.

Assume that the relation (3.3) is true for \z
{n=r}. Then
\begin{multline*}
    \sum_{i=0}^{ 2(r+1)}\,J_{2k+2i+1}^2=
J_{2k+4r+5}^2+J_{2k+4r+3}^2+
\sum_{i=0}^{2r}\,J_{2k+2i+1}^2=
J_{2k+4r+5}^2+J_{2k+4r+3}^2+
\sum_{i=0}^{2r}\,J_{2i+1}^2\\+8\,{\beta}_n\,J_{2\,k}
\left[2\,{\gamma}_n\,J_{2k-2}+ {\pi}_n\right]
=\sum_{i=0}^{2(r+1)}\,J_{2i+1}^2+8\,{\beta}_{n+1}\,
J_{2\,k}\left[2\,{\gamma}_{n+1}\,J_{2k-2}+
{\pi}_{n+1}\right],
\end{multline*}
where the last step uses Lemma 5 for \z
{n=r+1}.
\end{proof}
\begin{proof}[Proof of (3.4)] ({\textbf{P3}}).
The proof is by induction on \z {n}. When \z
{n=0} we obtain \q {
    J_{2k+1}^2+J_{2k+3}^2=
    J_{1}^2+J_{3}^2+
8\,{\tau}_{1}\,J_{2\,k}
\left[2\,{\eta}_0\,J_{2k-2}+{\sigma}_0\right]
= 10+8\,J_{2\,k}
\left[34\,J_{2k-2}+15\right],} since \z
{J_1=1}, \z {J_3=3}, \z {{\tau}_1=1}, \z
{{\eta}_0=17} and \z {{\sigma}_0=15}. The
above equality is true by Lemma 4.

Assume that the relation (3.4) is true for \z
{n=r}. Then
\begin{multline*}
    \sum_{i=0}^{ 2(r+1)+1}\,J_{2k+2i+1}^2=
J_{2k+4r+7}^2+J_{2k+4r+5}^2+
\sum_{i=0}^{2r+1}\,J_{2k+2i+1}^2=\\
J_{2k+4r+7}^2+J_{2k+4r+5}^2+
\sum_{i=0}^{2r+1}\,J_{2i+1}^2+8\,{\tau}_{r+1}\,
J_{2\,k}\left[2\,{\eta}_r\,J_{2k-2}+ {\sigma}_r\right]\\
=\sum_{i=0}^{2(r+1)+1}\,J_{2i+1}^2+8\,{\tau}_{r+2}\,
J_{2\,k}\left[2\,{\eta}_{r+1}\,J_{2k-2}+
{\sigma}_{r+1}\right],
\end{multline*}
where the last step uses Lemma 6 for \z
{n=r+1}.
\end{proof}


\section{Jacobsthal products}
For the first two steps in a proof by
induction of our next theorem we require the
following lemma.

\begin{lem}\label{l7}
For every \z {k\geqslant 0} the following
equalities hold \qq {J_{2k+1}=8\,J_{2k-2}+3.}
\qq {J_{2k}\,J_{2k+1}+J_{2k+2}\,J_{2k+3}=
3+J_{2k}\,(136\,J_{2k-2}+55).}
\end{lem}
\begin{proof}[Proof of (4.1)]
    By the formula \z {J_k={\frac{2^k-(-1)^k}{3}}} we have
    \begin{multline*}
    J_{2k+1}={\frac{2^{2k+1}-(-1)^{2k+1}}{3}}=
{\frac{2^{2k+1}+1}{3}}
=8\cdot{\frac{2^{2k-2}-(-1)^{2k-2}}{3}}
+3=8\,J_{2k-2}+3.
\end{multline*}
\end{proof}
\begin{proof}[Proof of (4.2)] ({\textbf{P2}}).
\z {{\frac{\left( P-1 \right) \left( P+1
\right)}{3}}\left(55\,M^2-46\,{P}^{2}+9
\right).}
\end{proof}

With the following lemma we shall make the
inductive step in the proof of the first part
of our third theorem.
\begin{lem}\label{l8}
For every \z {m\geqslant 0} and \z {k\geqslant
0} the following equality holds
\begin{multline} \hspace{2cm}
J_{2k}\,\left[8\,({\beta}_{n+1}\,
   {\gamma}_{n+1}-{\beta}_{n}\,{\gamma}_{n})\,
   J_{2k-2}+
{\beta}_{n+1}\,{\mu}_{n+1}-
{\beta}_{n}\,{\mu}_{n}\right]=\\
J_{2k+4n+5}\,J_{2k+4n+4}+J_{2k+4n+3}\,J_{2k+4n+2}-
J_{4n+5}\,J_{4n+4}-J_{4n+3}\,J_{4n+2}.
\hspace{1cm}
\end{multline}
\end{lem}
\begin{proof} ({\textbf{P1}}).
\begin{multline*}
 {\frac
{2720}{9}}\,\left(  \left( -1 \right)
^{1+2\,k}+1 \right) AB+{\frac {20}{9}}\,
\left( \left( -1 \right) ^{4\,n+2\,k}+ \left(
-1 \right) ^{4\,n+1} \right) (A+1)C+\\ { \frac
{544}{9}}\,\left( 4\,\left( -1 \right)
^{4\,k}+5\, \left( -1 \right)
^{1+2\,k}+1\right) B+{\frac{2}{9}}\,\left(
 \left( -1 \right) ^{8\,n+1}+
 \left( -1 \right) ^{8\,n+4\,k}\right).
\end{multline*}
\end{proof}
With the following lemma we shall make the
inductive step in the proof of the second part
of our third theorem.
\begin{lem}\label{l9}
For every \z {m\geqslant 0} and \z {k\geqslant
0} the following equality holds
\begin{multline}
\hspace{1.3cm}
J_{2k}\,\left[8\,({\tau}_{n+2}\,
   {\eta}_{n+1}-{\tau}_{n+1}\,{\eta}_{n})\,
   J_{2k-2}+
{\tau}_{n+2}\,{\nu}_{n+1}-
{\tau}_{n+1}\,{\nu}_{n}\right]=\\
J_{2k+4n+5}\,J_{2k+4n+4}+J_{2k+4n+7}\,J_{2k+4n+6}-
J_{4n+5}\,J_{4n+4}-J_{4n+7}\,J_{4n+6}.
\hspace{1cm}
\end{multline}
\end{lem}
\begin{proof} ({\textbf{P1}}).
\begin{multline*}
{\frac {43520}{9}}\,\left(  \left( -1 \right)
^{1+2\,k}+1 \right) AB+ {\frac {80}{9}}\,
\left( \left( -1 \right) ^{4\,n+2\,k}+ \left(
-1 \right) ^{4\,n+1} \right)
\left(A+1\right)C\\+ { \frac
{8704}{9}}\,\left( 4\,\left( -1 \right)
^{4\,k}+5\, \left( -1 \right)
^{1+2\,k}+1\right) B+{\frac{2}{9}}\,\left(
 \left( -1 \right) ^{8\,n+1}+
 \left( -1 \right) ^{8\,n+4\,k}\right).
\end{multline*}
\end{proof}
For any \z {n=0,\,1,\,2,\dots} let \z
{{\mu}_n={\frac{2^{4n+3}+1}{3}}=J_{4n+3}} and
\z
{{\nu}_n={\frac{5\,\left(2^{4n+5}+1\right)}{3}}}
\z {=5\,J_{4n+5}}.
\begin{thm}\label{t3}
For every \z {m\geqslant 0} and \z {k\geqslant
0} the following equalities hold \qq
{\sum_{i=0}^{m}\,J_{2k+2i}\,J_{2k+2i+1}=
\sum_{i=0}^{m}\,J_{2i}\,J_{2i+1}+{\beta}_n\,J_{2\,k}
\left[8\,{\gamma}_n\,J_{2k-2}+
{\mu}_n\right],} if \z {m=2n} and \z
{n=0,\,1,\,2,\dots}  and \qq {
\sum_{i=0}^{m}\,J_{2k+2i}\,J_{2k+2i+1}=
\sum_{i=0}^{m}\,J_{2i}\,J_{2i+1}+{\tau}_{n+1}\,J_{2\,k}
\left[8\,{\eta}_n\,J_{2k-2}+{\nu}_n\right],}
if \z {m=2n+1} and \z {n=0,\,1,\,2,\dots}.
\end{thm}

\begin{proof}[Proof of (4.5)] ({\textbf{P3}}).
For \z {n=0} the relation (4.5) is \q
{J_{2k}\,J_{2k+1}=J_0\,J_1+{\beta}_0\,J_{2k}(
8\,{\gamma}_0\,J_{2k-2}+{\mu}_0)=J_{2k}
(8\,J_{2k-2}+3)} which is true since \z
{J_{2k+1}=8\,J_{2k-2}+3} by the relation (4.1)
in Lemma~7.

Assume that the relation (4.5) is true for \z
{n=r}. Then
\begin{multline*}
    \sum_{i=0}^{ 2(r+1)}\,J_{2k+2i}\,J_{2k+2i+1}=
J_{2k+4r+4}\,J_{2k+4r+5}+J_{2k+4r+2}\,J_{2k+4r+3}
 +\sum_{i=0}^{2r}\,J_{2k+2i}\,J_{2k+2i+1}=\\
J_{2k+4r+4}\,J_{2k+4r+5}+J_{2k+4r+2}\,J_{2k+4r+3}
+\sum_{i=0}^{2r}\,J_{2i}\,J_{2i+1}+{\beta}_n\,J_{2\,k}
\left[8\,{\gamma}_n\,J_{2k-2}+ {\mu}_n\right]\\
=\sum_{i=0}^{2(r+1)}\,J_{2i}\,J_{2i+1}+{\beta}_{n+1}\,
J_{2\,k}\left[8\,{\gamma}_{n+1}\,J_{2k-2}+
{\mu}_{n+1}\right],
\end{multline*}
where the last step uses Lemma 8 for \z
{n=r+1}.
\end{proof}
\begin{proof}[Proof of (4.6)] ({\textbf{P3}}).
For \z {n=0} the relation (4.6) is
\begin{multline*}
    J_{2k}\,J_{2k+1}+J_{2k+2}\,J_{2k+3}=J_0\,J_1+
J_2\,J_3+{\tau}_1\,J_{2k}(
8\,{\eta}_0\,J_{2k-2}+{\nu}_0)
    =3+J_{2k}
(136\,J_{2k-2}+55)
\end{multline*}
which is true by (4.2) in Lemma 7.

Assume that the relation (4.6) is true for \z
{n=r}. Then
\begin{multline*}
    \sum_{i=0}^{ 2(r+1)+1}\,J_{2k+2i}\,J_{2k+2i+1}=
J_{2k+4r+4}\,J_{2k+4r+5}+J_{2k+4r+6}\,J_{2k+4r+7}
 +\sum_{i=0}^{2r+1}\,J_{2k+2i}\,J_{2k+2i+1}=\\
J_{2k+4r+4}\,J_{2k+4r+5}+J_{2k+4r+6}\,J_{2k+4r+7}
+\sum_{i=0}^{2r+1}\,J_{2i}\,J_{2i+1}+{\tau}_{n+1}\,J_{2\,k}
\left[8\,{\eta}_n\,J_{2k-2}+ {\nu}_n\right]\\
=\sum_{i=0}^{2(r+1)+1}\,J_{2i}\,J_{2i+1}+
{\tau}_{n+2}\,
J_{2\,k}\left[8\,{\eta}_{n+1}\,J_{2k-2}+
{\nu}_{n+1}\right],
\end{multline*}
where the last step uses Lemma 9 for \z
{n=r+1}.
\end{proof}

\section{Alternating Jacobsthal even squares}
In this section we look for formulas that give
closed forms for alternating sums of squares
of Jacobsthal numbers with even indices.
\begin{lem}\label{l10}
For every \z {k\geqslant 0} we have \qq
{J_{2k}=4\,J_{2k-2}+1.}
\end{lem}
\begin{proof}
By the formula \z {J_k={\frac{2^k-(-1)^k}{3}}}
we get \q {
4\,J_{2k-2}+1=4\cdot{\frac{2^{2k-2}-(-1)^{2k-2}}{3}}+
1={\frac{2^{2k}-(-1)^{2k}}{3}}=J_{2k}.}
\end{proof}


\begin{lem}\label{l11}
    For every \z {k\geqslant 0} we have
    \qq {J_{2k+2}^2-J_{2k}^2=
    1+J_{2k}\,(60\,J_{2k-2}+23).}
\end{lem}
\begin{proof} ({\textbf{P2}}).
\z {{\frac{\left( P-1 \right) \left( P+1
\right)}{3}}\left(23\,M^2-20\,{P}^{2}+3
\right).}
\end{proof}
Let \z {{\tau}_0^*=1} and \z
{{\tau}_{n+1}^*-{\tau}_{n}^*=
2^{4n+3}\,(50\cdot 2^{4n}-1),} for \z
{n=0,\,1,\,2,\dots}. For any \z
{n=0,\,1,\,2,\dots} let \z
{{\gamma}_n^*={\frac{2^{8n+4}+1}{17}}=
{\frac{j_{8n+4}}{17}}}.
\begin{lem}\label{l12}
    For every \z {k\geqslant 0} and every
    \z {n\geqslant 0} we have
\begin{multline}
    \hspace{1cm}J_{2k}\,(4\,({\gamma}_{n+1}^*-
    {\gamma}_{n}^*)\,J_{2k-2}+{\tau}_{n+1}^*-
    {\tau}_{n}^*)=J_{2k+4n+4}^2-J_{2k+4n+2}^2-
    J_{4n+4}^2+J_{4n+2}^2.\hspace{0.6cm}
        \end{multline}
    \end{lem}
\begin{proof} ({\textbf{P1}}).
\begin{multline*}
{\frac {400}{3}}\,\left(  \left( -1 \right)
^{1+2\,k}+1 \right) AB+ {\frac {8}{3}}\,
\left( \left( -1 \right) ^{4\,n+2\,k} -1
\right) AC+\\{ \frac {80}{3}}\,\left(
4\,\left( -1 \right) ^{4\,k}+5\, \left( -1
\right) ^{1+2\,k}+1\right) B +{\frac {8}{3}
}\,\left(\left( -1 \right) ^{2\,k}+\left( -1
 \right) ^{4\,n+1} \right) C.
\end{multline*}
\end{proof}
Let \z {{\beta}_0^*=23} and \z
{{\beta}_{n+1}^*-{\beta}_{n}^*=2^{4n+5}\left(200\cdot
2^{4n}-1\right)} for any \z
{n=0,\,1,\,2,\dots}. For the same values of \z
{n}, let \z
{{\eta}_n^*={\frac{2^{8n+8}-1}{17}}={\frac{3}{17}}\,
J_{8n+8}}.
\begin{lem}\label{l13}
    For every \z {k\geqslant 0} and every
    \z {n\geqslant 0} we have
\begin{multline}
    \hspace{1cm}J_{2k}\,(4\,({\eta}_{n+1}^*-
    {\eta}_{n}^*)\,J_{2k-2}+{\beta}_{n+1}^*-
    {\beta}_{n}^*)=J_{2k+4n+6}^2-J_{2k+4n+4}^2+
   J_{4n+4}^2-J_{4n+6}^2.\hspace{0.5cm}
        \end{multline}
\end{lem}
\begin{proof} ({\textbf{P1}}).
\begin{multline*}
{\frac {6400}{3}}\,\left(  \left( -1 \right)
^{1+2\,k}+1 \right) AB+ {\frac {32}{3}}\,
\left( \left( -1 \right) ^{4\,n+2\,k} -1
\right) AC+\\{ \frac {1280}{3}}\,\left(
4\,\left( -1 \right) ^{4\,k}+5\, \left( -1
\right) ^{1+2\,k}+1\right) B +{\frac {32}{3}
}\,\left(\left( -1 \right) ^{2\,k}+\left( -1
 \right) ^{4\,n+1} \right) C.
\end{multline*}
\end{proof}
\begin{thm}\label{t4}
For every \z {m\geqslant 0} and \z {k\geqslant
0} the following equalities hold \qq
{\sum_{i=0}^{m}\,(-1)^i\,J_{2k+2i}^2=
\sum_{i=0}^{m}\,(-1)^i\,J_{2i}^2+J_{2\,k}
\left[4\,{\gamma}^*_n\,J_{2k-2}+
{\tau}^*_n\right],} if \z {m=2n} and \z
{n=0,\,1,\,2,\dots}  and \qq {
\sum_{i=0}^{m}\,(-1)^i\,J_{2k+2i}^2=
\sum_{i=0}^{m}\,(-1)^i\,J_{2i}^2-J_{2\,k}
\left[4\,{\eta}^*_n\,J_{2k-2}+
{\beta}^*_{n}\right],} if \z {m=2n+1} and \z
{n=0,\,1,\,2,\dots}.
\end{thm}

\begin{proof}[Proof of (5.5)] ({\textbf{P3}}).
For \z {n=0} the relation (5.5) is \q
{J_{2k}^2=J_{0}^2+J_{2k}\,(4\,{\gamma}_0^*\,
J_{2k-2}+{\tau}_0^*)=J_{2k}\,(4\, J_{2k-2}+1)}
(i. e., the relation (5.1) multiplied by \z
{J_{2k}}) which is true by Lemma 10.

Assume that the relation (5.5) is true for \z
{n=r}. Then
\begin{multline*}
    \sum_{i=0}^{2(r+1)}\,(-1)^i\cdot
J_{2k+2i}^2=Q_{2k+4r+4}^2-Q_{2k+4r+2}^2+
\sum_{i=0}^{2r}\,(-1)^i\cdot
J_{2k+2i}^2\\
=Q_{2k+4r+4}^2-Q_{2k+4r+2}^2+
\sum_{i=0}^{2r}\,(-1)^i\cdot J_{2i}^2+
J_{2\,k} \left[4\,{\gamma}^*_n\,J_{2k-2}+
{\tau}^*_n\right]=\\
\sum_{i=0}^{2(r+1)}\,(-1)^i\cdot J_{2i}^2+
J_{2\,k} \left[4\,{\gamma}^*_{n+1}\,J_{2k-2}+
{\tau}^*_{n+1}\right],
\end{multline*}
where the last step uses Lemma 12.
\end{proof}

\begin{proof}[Proof of (5.6)] ({\textbf{P3}}).
For \z {n=0} the relation (5.6) is \q {
    J_{2k}^2-J_{2k+2}^2=(J_0^2-J_2^2)-
J_{2\,k} \left[4\,{\eta}^*_0\,J_{2k-2}+
{\beta}^*_{0}\right]
    =-1-J_{2\,k}
\left[60\,J_{2k-2}+ 23\right],} which is true
by Lemma 11.

Assume that the relation (5.6) is true for \z
{n=r}. Then
\begin{multline*}
    \sum_{i=0}^{2(r+1)+1}\,(-1)^i\cdot
J_{2k+2i}^2=J_{2k+4r+4}^2-J_{2k+4r+6}^2+
\sum_{i=0}^{2r+1}\,(-1)^i\cdot
J_{2k+2i}^2\\
=J_{2k+4r+4}^2-J_{2k+4r+6}^2+
\sum_{i=0}^{2r+1}\,(-1)^i\cdot J_{2i}^2-
J_{2\,k} \left[4\,{\eta}^*_n\,J_{2k-2}+
{\beta}^*_{n}\right]\\
=\sum_{i=0}^{2(r+1)+1}\,(-1)^i\cdot J_{2i}^2-
J_{2\,k} \left[4\,{\eta}^*_{n+1}\,J_{2k-2}+
{\beta}^*_{n+1}\right],
\end{multline*}
where the last step uses Lemma 13.
\end{proof}

\section{Alternating Jacobsthal odd squares}
In this section we look for formulas that give
closed forms for alternating sums of squares
of Jacobsthal numbers with odd indices.
\begin{lem}\label{l14}
For every \z {k\geqslant 0} we have \qq
{J_{2k+1}^2=1+8\,J_{2k}\left[
2\,J_{2k-2}+1\right].}
\end{lem}
\begin{proof} ({\textbf{P2}}).
\z {{\frac{\left( P-1 \right) \left( P+1
\right)}{3}}\left(8\,M^2-5\,{P}^{2}+3
\right).}
\end{proof}

\begin{lem}\label{l15}
    For every \z {k\geqslant 0} we have
    \qq {J_{2k+3}^2-J_{2k+1}^2=
    8+8\,J_{2k}\,(30\,J_{2k-2}+13).}
\end{lem}
\begin{proof} ({\textbf{P2}}).
\z {{\frac{\left( P-1 \right) \left( P+1
\right)}{3}}\left(13\,M^2-10\,{P}^{2}+3
\right).}
\end{proof}
Let \z {{\tau}_0^{**}=1} and \z
{{\tau}_{n+1}^{**}-{\tau}_{n}^{**}= 101\cdot
2^{4n+1}+25\cdot 2^{4n+3}\,(2^{4n}-1)} for \z
{n=0,\,1,\,2,\dots}.
\begin{lem}\label{l16}
    For every \z {k\geqslant 0} and every
    \z {n\geqslant 0} we have
\begin{multline}
8\,J_{2k}\,(2\,({\gamma}_{n+1}^*-
    {\gamma}_{n}^*)\,J_{2k-2}+{\tau}_{n+1}^{**}-
    {\tau}_{n}^{**})=
J_{2k+4n+5}^2-J_{2k+4n+3}^2-
    J_{4n+5}^2+J_{4n+3}^2.\hspace{0.5cm}
        \end{multline}
    \end{lem}
\begin{proof} ({\textbf{P1}}).
\begin{multline*}
{\frac {1600}{3}}\,\left(  \left( -1 \right)
^{1+2\,k}+1 \right) AB+ {\frac {16}{3}}\,
\left( \left( -1 \right) ^{4\,n+2\,k+1} +1
\right) AC+\\{ \frac {320}{3}}\,\left(
4\,\left( -1 \right) ^{4\,k}+5\, \left( -1
\right) ^{1+2\,k}+1\right) B +{\frac {16}{3}
}\,\left(\left( -1 \right) ^{2\,k+1}+\left( -1
 \right) ^{4\,n} \right) C.
\end{multline*}
\end{proof}
%For any \z {n=0,\,1,\,2,\dots} let \z
%{{\eta}_n^*={\frac{2^{8n+8}-1}{17}}={\frac{3}{17}}\,
%J_{8n+8}}.

Let \z {{\beta}_0^{**}=13} and \z
{{\beta}_{n+1}^{**}-{\beta}_{n}^{**}= 401\cdot
2^{4n+3}+100\cdot
2^{4n+5}\left(2^{4n}-1\right)} for any \z
{n=0,\,1,\,2,\dots}.
\begin{lem}\label{l17}
    For every \z {k\geqslant 0} and every
    \z {n\geqslant 0} we have
\begin{multline}
8\,J_{2k}\,(2\,({\eta}_{n+1}^*-
    {\eta}_{n}^*)\,J_{2k-2}+{\beta}_{n+1}^{**}-
    {\beta}_{n}^{**})=
J_{2k+4n+7}^2-J_{2k+4r+5}^2+
   J_{4n+5}^2-J_{4r+7}^2.\hspace{0.8cm}
        \end{multline}
\end{lem}
\begin{proof} ({\textbf{P1}}).
\begin{multline*}
{\frac {25600}{3}}\,\left(  \left( -1 \right)
^{1+2\,k}+1 \right) AB+ {\frac {64}{3}}\,
\left( \left( -1 \right) ^{4\,n+2\,k+1} +1
\right) AC+\\{ \frac {5120}{3}}\,\left(
4\,\left( -1 \right) ^{4\,k}+5\, \left( -1
\right) ^{1+2\,k}+1\right) B +{\frac {64}{3}
}\,\left(\left( -1 \right) ^{2\,k+1}+\left( -1
 \right) ^{4\,n} \right) C.
\end{multline*}
\end{proof}
\begin{thm}\label{t5}
For every \z {m\geqslant 0} and \z {k\geqslant
0} the following equalities hold \qq
{\sum_{i=0}^{m}\,(-1)^i\,J_{2k+2i+1}^2=
\sum_{i=0}^{m}\,(-1)^i\,J_{2i+1}^2+8\,J_{2\,k}
\left[2\,{\gamma}^*_n\,J_{2k-2}+
{\tau}^{**}_n\right],} if \z {m=2n} and \z
{n=0,\,1,\,2,\dots}  and \qq {
\sum_{i=0}^{m}\,(-1)^i\,J_{2k+2i+1}^2=
\sum_{i=0}^{m}\,(-1)^i\,J_{2i+1}^2-8\,J_{2\,k}
\left[2\,{\eta}^*_n\,J_{2k-2}+
{\beta}^{**}_n\right],} if \z {m=2n+1} and \z
{n=0,\,1,\,2,\dots}.
\end{thm}

\begin{proof}[Proof of (6.5)] ({\textbf{P3}}).
For \z {n=0} the relation (6.5) is \q
{J_{2k+1}^2=J_1^2+8\,J_{2k}(2\,{\gamma}_0^*\,
J_{2k-2}+{\tau}_0^{**})} (i. e., the relation
(6.1)) which is true by Lemma 14.

Assume that the relation (6.5) is true for \z
{n=r}. Then
\begin{multline*}
    \sum_{i=0}^{2(r+1)}\,(-1)^i\,
J_{2k+2i+1}^2=J_{2k+4r+5}^2-J_{2k+4r+3}^2+
\sum_{i=0}^{2r}\,(-1)^i\,
J_{2k+2i+1}^2\\
=J_{2k+4r+5}^2-J_{2k+4r+3}^2+
\sum_{i=0}^{2r}\,(-1)^i\,
J_{2i+1}^2+8\,J_{2\,k}
\left[2\,{\gamma}^*_n\,J_{2k-2}+
{\tau}^{**}_n\right]\\
=\sum_{i=0}^{2(r+1)}\,(-1)^i\,
J_{2i+1}^2+8\,J_{2\,k}
\left[2\,{\gamma}^*_{n+1}\,J_{2k-2}+
{\tau}^{**}_{n+1}\right],
\end{multline*}
where the last step uses Lemma 16.
\end{proof}

\begin{proof}[Proof of (6.6)] ({\textbf{P3}}).
The proof is again by induction on \z {n}. For
\z {n=0} the relation (6.6) is \z
{J_{2k+1}^2-J_{2k+3}^2=J_1^2-J_3^2-
8\,J_{2\,k} \left[2\,{\eta}^*_0\,J_{2k-2}+
{\beta}^{**}_0\right]} which is true by Lemma
15 since \z {{\eta}^*_0=15} and \z
{{\beta}^{**}_0=13}.

Assume that the relation (6.6) is true for \z
{n=r}. Then
\begin{multline*}
 \sum_{i=0}^{2(r+1)+1}\,(-1)^i\,
J_{2k+2i+1}^2=J_{2k+4r+5}^2-J_{2k+4r+7}^2+
\sum_{i=0}^{2r+1}\,(-1)^i\, J_{2k+2i+1}^2\\
=J_{2k+4r+5}^2-J_{2k+4r+7}^2+
\sum_{i=0}^{2r+1}\,(-1)^i\,
J_{2i+1}^2-8\,J_{2\,k}
\left[2\,{\eta}^*_n\,J_{2k-2}+
{\beta}^{**}_n\right]\\
=\sum_{i=0}^{2(r+1)+1}\,(-1)^i\,
J_{2i+1}^2-8\,J_{2\,k}
\left[2\,{\eta}^*_{n+1}\,J_{2k-2}+
{\beta}^{**}_{n+1}\right],
\end{multline*}
where the last step uses Lemma 17.
\end{proof}


\section{Alternating Jacobsthal products}

\begin{lem}\label{l18}
    For every \z {k\geqslant 0} we have
    \qq {J_{2k+3}\,J_{2k+2}-
    J_{2k+1}\,J_{2k}=3+J_{2k}\,\left[
    120\,J_{2k-2}+49\right].}
\end{lem}
\begin{proof} ({\textbf{P2}}).
\z {{\frac{\left( P-1 \right) \left( P+1
\right)}{3}}\left(49\,M^2-40\,{P}^{2}+9
\right).}
\end{proof}
\begin{lem}\label{l19}
    For every \z {k\geqslant 0} and every
    \z {n\geqslant 0} we have
\begin{multline}
\hspace{1cm}
J_{2k}\,\left(8\,({\gamma}_{n+1}^*-
    {\gamma}_{n}^*)\,J_{2k-2}+{\tau}_{n+1}^{***}-
    {\tau}_{n}^{***}\right)=\\
   J_{2k+4n+5}\,J_{2k+4n+4}-
   J_{2k+4n+3}\,J_{2k+4n+2}-J_{4n+5}\,J_{4n+4}
   +J_{4n+3}\,J_{4n+2}.
   \hspace{1cm}
        \end{multline}
    \end{lem}
\begin{proof} ({\textbf{P1}}).
\begin{multline*}
{\frac {800}{3}}\,\left(  \left( -1 \right)
^{1+2\,k}+1 \right) AB+ {\frac {4}{3}}\,
\left( \left( -1 \right) ^{4\,n+2\,k} -1
\right) AC+\\{ \frac {160}{3}}\,\left(
4\,\left( -1 \right) ^{4\,k}+5\, \left( -1
\right) ^{1+2\,k}+1\right) B +{\frac {4}{3}
}\,\left(\left( -1 \right) ^{2\,k}+\left( -1
 \right) ^{4\,n+1} \right) C.
\end{multline*}
\end{proof}
Let \z {{\beta}_0^{***}=49} and \z
{{\beta}_{n+1}^{***}-{\beta}_{n}^{***}=
799\cdot 2^{4n+4}+25\cdot
2^{4n+9}\left(2^{4n}-1\right)} for any \z
{n=0,\,1,\,2,\dots}.
\begin{lem}\label{l20}
    For every \z {k\geqslant 0} and every
    \z {n\geqslant 0} we have
\begin{multline}
\hspace{1cm}J_{2k}\,(8\,({\eta}_{n+1}^*-
    {\eta}_{n}^*)\,J_{2k-2}+{\beta}_{n+1}^{***}-
    {\beta}_{n}^{***})=\\
   J_{2k+4n+6}\,J_{2k+4n+7}-
   J_{2k+4n+4}\,J_{2k+4n+5}+J_{4n+5}\,J_{4n+4}
   -J_{4n+6}\,J_{4n+7}.\hspace{1cm}
        \end{multline}
\end{lem}
\begin{proof} ({\textbf{P1}}).
\begin{multline*}
{\frac {12800}{3}}\,\left(  \left( -1 \right)
^{1+2\,k}+1 \right) AB+ {\frac {16}{3}}\,
\left( \left( -1 \right) ^{4\,n+2\,k} -1
\right) AC+\\{ \frac {2560}{3}}\,\left(
4\,\left( -1 \right) ^{4\,k}+5\, \left( -1
\right) ^{1+2\,k}+1\right) B +{\frac {16}{3}
}\,\left(\left( -1 \right) ^{2\,k}+\left( -1
 \right) ^{4\,n+1} \right) C.
\end{multline*}
\end{proof}

\begin{thm}\label{t6}
For every \z {m\geqslant 0} and \z {k\geqslant
0} the following equalities hold \qq {
    \sum_{i=0}^{m}\,(-1)^i\,J_{2k+2i}\,J_{2k+2i+1}=
    \sum_{i=0}^{m}\,(-1)^i\,J_{2i}\,J_{2i+1}+
    J_{2\,k}\left[8\,{\gamma}^*_n\,J_{2k-2}+
{\tau}^{***}_n\right],} if \z {m=2n} and \z
{n=0,\,1,\,2,\dots}  and \qq {
\sum_{i=0}^{m}\,(-1)^i\,J_{2k+2i}\,J_{2k+2i+1}=
\sum_{i=0}^{m}\,(-1)^i\,J_{2i}\,J_{2i+1}-J_{2\,k}
\left[8\,{\eta}^*_n\,J_{2k-2}+
{\beta}^{***}_n\right],} if \z {m=2n+1} and \z
{n=0,\,1,\,2,\dots}.
\end{thm}

\begin{proof}[Proof of (7.4)] ({\textbf{P3}}).
For \z {n=0} the relation (7.4) is \q
{J_{2k}\,J_{2k+1}=J_0\,J_1+
    J_{2\,k}\left[8\,{\gamma}^*_0\,J_{2k-2}+
{\tau}^{***}_0\right]=
J_{2\,k}\left[8\,J_{2k-2}+3\right]} which is
true by the relation (4.1) in Lemma 7 since \z
{J_0=0}, \z {J_1=1}, \z {{\gamma}^*_0=1} and
\z {{\tau}^{***}_0=3}.

Assume that the relation (7.4) is true for \z
{n=r}. Then
\begin{multline*}
\hspace{-0.2cm}\sum_{i=0}^{2(r+1)}(-1)^i
J_{2k+2i}J_{2k+2i+1}=\sum_{i=0}^{2r}(-1)^i
J_{2k+2i}J_{2k+2i+1}+ J_{2k+4r+4}J_{2k+4r+5}-
J_{2k+4r+2}J_{2k+4r+3}\\
=J_{2k+4r+4}J_{2k+4r+5}-
J_{2k+4r+2}J_{2k+4r+3}+\sum_{i=0}^{2r}(-1)^i
J_{2i}J_{2i+1} +J_{2\,k}
\left[8\,{\gamma}^*_n\,J_{2k-2}+
{\tau}^{***}_n\right]\\
=\sum_{i=0}^{2(r+1)}(-1)^i J_{2i}J_{2i+1}
+J_{2\,k} \left[8\,{\gamma}^*_{n+1}\,J_{2k-2}+
{\tau}^{***}_{n+1}\right],
\end{multline*}
where the last step uses Lemma 19.
\end{proof}

\begin{proof}[Proof of (7.5)] ({\textbf{P3}}).
For \z {n=0} the relation (7.5) is
\begin{multline*}
    J_{2k}\,J_{2k+1}-J_{2k+2}\,J_{2k+3}=
    J_0\,J_1-
J_2\,J_3-\\
    J_{2k}\left[8\,{\eta}^*_0\,J_{2k-2}
+{\beta}^{***}_0\right] =-3-J_{2k}
\left[120\,J_{2k-2} +49\right],
\end{multline*}
(i. e., the relation (7.1)) which is true by
Lemma 18.

Assume that the relation (7.5) is true for \z
{n=r}. Then
\begin{multline*}
    \sum_{i=0}^{2(r+1)+1}\,(-1)^i\,
J_{2k+2i}\,J_{2k+2i+1}=\sum_{i=0}^{2r+1}\,(-1)^i\,
J_{2k+2i}\,J_{2k+2i+1}+
J_{2k+4r+4}\,J_{2k+4r+5}-\\
J_{2k+4r+6}\,J_{2k+4r+7}
=J_{2k+4r+4}\,J_{2k+4r+5}-
J_{2k+4r+6}\,J_{2k+4r+7}+\sum_{i=0}^{2r+1}\,(-1)^i\,
J_{2i}\,J_{2i+1} -\\ J_{2\,k}
\left[8\,{\eta}^*_n\,J_{2k-2}+
{\beta}^{***}_n\right]
=\sum_{i=0}^{2(r+1)+1}\,(-1)^i\,
J_{2i}\,J_{2i+1} -J_{2\,k}
\left[8\,{\eta}^*_{n+1}\,J_{2k-2}+
{\beta}^{***}_{n+1}\right],
\end{multline*}
where the last step uses Lemma 20.
\end{proof}








\bigskip
\hrule
\bigskip

\noindent 2000 {\it Mathematics Subject
Classification}: Primary 11B39; Secondary 11Y55, 05A19.

\noindent \emph{Keywords:} Jacobsthal numbers,
Jacobsthal-Lucas numbers, integer sequences,
sum of squares, combinatorial identity, Maple.



\bigskip
\hrule
\bigskip

\noindent (Concerned with sequences
\seqnum{A001045} and \seqnum{A014551}.)

\bigskip
\hrule
\bigskip


\vspace*{+.1in}
\noindent Received May 17 2006; revised version received January 19 2007.
Published in {\it Journal of Integer Sequences}, January 19 2007.

\bigskip
\hrule
\bigskip

\noindent Return to \htmladdnormallink{Journal
of Integer Sequences home
page}{http://www.math.uwaterloo.ca/JIS/}.

\end{document}
